Author’s response to reviews

Title: Fostering patient-centredness by following patients outside the clinical setting: an interview study

Authors:
Christel Canét-Wittkampf (C.T.M.Wittkampf@umcutrecht.nl)
Charlotte Eijkelboom (M.C.L.Eijkelboom@umcutrecht.nl)
Saskia Mol (s.s.l.mol@umcutrecht.nl)
Dorien Zwart (d.zwart@umcutrecht.nl)
Iris Hendriks (i.hendriks@students.uu.nl)
Esther de Groot (E.deGroot@umcutrecht.nl)

Version: 2 Date: 19 Dec 2019

Author’s response to reviews:

Dear editor,

Attached you will find our manuscript in which minor revisions have been made.

In the first step it was recommended to us (by Liam, the previous editor) that I would submit the paper because Christel and Charlotte both are first authors on this paper. Therefore, I have been the corresponding author up till now. I will change it in the system.

Best wishes, Esther de Groot